Author’s response to reviews

Title: Predictors of Exclusive Breast Feeding: Observations from the Alberta Pregnancy Outcomes and Nutrition (APrON) study

Authors:

Mahsa Jessri (jessri@ualberta.ca)
Anna P Farmer (Anna.farmer@ualberta.ca)
Katerina Maximova (katerina.maximova@ualberta.ca)
Noreen D Willows (Noreen.Willows@ales.ualberta.ca)
Rhonda C Bell (Rhonda.Bell@ales.ualberta.ca)

Version: 2 Date: 24 February 2013

Author’s response to reviews: see over
Thank you for providing us an opportunity to revise the manuscript entitled “Predictors of Exclusive Breast Feeding: Observations from the Alberta Pregnancy Outcomes and Nutrition (APrON) study” for publication in the “BMC Pediatrics”. We have taken each critique and comment very seriously, and have provided point-by-point response to the reviewer’s concerns. In addition the entire text of the manuscript has been reviewed, edited and improved. However, authors are willing to provide further modifications, in case it is needed.

Again we thank you for your consideration.

Sincerely,

A. Prof. Anna P Farmer, PhD, MPH, RD
Human Nutrition Division, Alberta Institute for Human Nutrition, and The Center for Health Promotion Studies, Edmonton Clinic Health Academy 4-370
University of Alberta, Edmonton, AB, Canada T6G 1C9

Email: Anna.farmer@ualberta.ca
Tel: (+1) 780 492 2693; Fax: (+1) 780 492 4265